**Author's response to reviews**

**Title:** Role of serum S100B and PET-CT in follow-up of patients with cutaneous melanoma B. Peric, I. Zagar, S Novakovic, J. Zgajnar, M. Hocevar Institute of Oncology Ljubljana, Slovenia

**Authors:**

Barbara Peric (bperic@onko-i.si)
Ivana Zagar (izagar@onko-i.si)
Srdjan Novakovic (snovakovic@onko-i.si)
Janez Zgajnar (jzgajnar@onko-i.si)
Marko Hocevar (mhocevar@onko-i.si)

**Version:** 4 **Date:** 20 July 2011

**Author's response to reviews:** see over
Dear Editor,

we are extremely pleased, that our manuscript will be published by your journal. This is a letter to confirm that following modifications were made to the manuscript:
   a) authors list was changed according to recommendations,
   b) the qualifications and name of the institution were removed from the title page,
   c) title Introduction was changed to Background,
   d) title Patients and Methods was changed to Methods,
   e) all visible vertical lines were removed from tables.

Sincerely,

Barbara Peric, MD, PhD